Author’s response to reviews

Title: Diet and nutrient status of legume consumers in Sweden: A descriptive cross-sectional study

Authors:

Céline Steib (celine.steib@lnu.se)

Ingegerd Johansson (ingegerd.johansson@umu.se)

Mohammed Hefni (mohammed.hefni@lnu.se)

Cornelia Witthöft (cornelia.witthoft@lnu.se)

Version: 2 Date: 25 Feb 2020

Author’s response to reviews:

Dear editor,

We would like to once again, thank the editor and the reviewers for taking time to review our manuscript.

We have addressed the comment of reviewer 1 in the revised manuscript. However, no comments were given from reviewer 3. After contacting the editorial managing team on two occasions, we have no gotten any answer from the reviewer. Therefore, we could not address the decision of minor revision from their part.

On behalf of all authors, With kind regards, Céline Steib

Reviewer 1:
Thank you for your time and comments. On line 327 we have added a sentence addressing the fact that the data does not represent usual intake: "Moreover, the data was not modeled for usual dietary intake and therefore only provides estimates."